
Daily status, June 5
SOAS ISFS Status, June 5, 2013

Tower work yesterday to fix TRH.8m and Tsfc.32m and clean optical sensors.

Also raised mote on tower to hopefully improve data transfer from rad and soil motes.

Sonics: 38m sonic has some flags in history

TRH, P, p: all okay; 

        problem with TRH.8m TRH was bad port #6 (& 17!) on low DSM

Wind speed & direction: ok

Soil, Rainr: OK!

        Vmote.(a,b,rad,pond) ~ 12.3, 12.2, 11.7, 12.3

Radiation, Wetness: 20m mote out from 05:45 to 10:00 CDT

                     8m mote out from 15:25 June 4

h2o, co2: 32m Licor has been flaky since hard rain on June 1.

System (GPS, Vdsm): ok 
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